Author's response to reviews

Title: Analysis of mutations in the E6 oncogene of Human Papillomavirus 16 in cervical cancer isolates from Moroccan women

Authors:

Zineb Qmichou (zineb.qmichou@gmail.com)
Meriem Khyatti (meriem.khyatti@pasteur.ma)
Mohamed Berraho (maberraho@yahoo.fr)
Moulay Mustapha Ennaji (m.ennaji@yahoo.fr)
Laila Benbacer (lbenbacer@yahoo.com)
Chakib Nejjari (cnejjari2000@yahoo.fr)
Noureddine Benjaafar (benjaafar@menara.ma)
Abdellatif Benider (beniderabdell@yahoo.fr)
Mohammed Attaleb (attaleb_mohammed@yahoo.fr)
Mohammed El Mzibri (mzibri@yahoo.com)

Version: 5 Date: 17 July 2013

Author's response to reviews:

Dear Professor,

Please find enclosed herewith, the manuscript untitled “Analysis of mutations in the E6 oncogene of Human Papillomavirus 16 in cervical cancer isolates from Moroccan women” revised as suggested by the referees. All modifications in the main text are made in blue color.

I would like to thank the referees for their relevant remarks and pertinent questions that have improved the submitted article.

You will find also point to point responses to all referees' comments and questions. I hope that we have corrected all the points raised in. If there is need to have more details, please let me know.

Sincerely yours